

Citation: Neuner-Jehle N, Zeeb M, Thorball CW, Fellay J, Metzner KJ, Frischknecht P, et al. (2024) Using viral diversity to identify HIV-1 variants under HLA-dependent selection in a systematic viral genome-wide screen. PLoS Pathog 20(8): e1012385. https://doi.org/10.1371/journal. ppat.1012385

Editor: Ronald Swanstrom, University of North Carolina at Chapel Hill, UNITED STATES OF AMERICA

Received: December 19, 2023

Accepted: July 2, 2024

Published: August 8, 2024

Peer Review History: PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.ppat.1012385

Copyright: © 2024 Neuner-Jehle et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data generated or analyzed during the current study cannot be **RESEARCH ARTICLE**

Using viral diversity to identify HIV-1 variants under HLA-dependent selection in a systematic viral genome-wide screen

Nadia Neuner-Jehle^{1,2}*, Marius Zeeb^{1,2}, Christian W. Thorball³, Jacques Fellay^{3,4,5}, Karin J. Metzner^{1,2}, Paul Frischknecht¹, Kathrin Neumann¹, Christine Leeman¹, Andri Rauch⁶, Marcel Stöckle⁷, Michael Huber², Matthieu Perreau⁸, Enos Bernasconi⁹, Julia Notter¹⁰, Matthias Hoffmann¹¹, Karoline Leuzinger¹², Huldrych F. Günthard^{1,2}, Chloé Pasin^{1,2,13°}, Roger D. Kouyos^{1,2°}, the Swiss HIV Cohort Study (SHCS)¹

 Department of Infectious Diseases and Hospital Epidemiology, University Hospital Zurich, Zurich, Switzerland, 2 Institute of Medical Virology, University of Zurich, Zurich, Switzerland, 3 Precision Medicine Unit, Biomedical Data Science Center, Lausanne University Hospital and University of Lausanne, Lausanne, Switzerland, 4 School of Life Sciences, École Polytechnique Fédérale de Lausanne, Lausanne, Switzerland, 5 Swiss Institute of Bioinformatics, Lausanne, Switzerland, 6 Department of Infectious Diseases, Bern University Hospital, University of Bern, Bern, Switzerland, 7 Division of Infectious Diseases and Hospital Epidemiology, University Hospital Basel, University of Basel, Basel, Switzerland, 8 Divisions of Immunology and Allergy, Lausanne University Hospital, Lausanne, Switzerland, 9 Division of Infectious Diseases, Ente Ospedaliero Cantonale, Lugano, University of Geneva and University of Southern Switzerland, Lugano, Switzerland, 10 Division of Infectious Diseases, Infection Prevention and Travel Medicine, Cantonal Hospital St. Gallen, St. Gallen, Switzerland, 11 Division of Infectious Diseases and Hospital Hospital Olten, Olten, Switzerland, 12 Clinical Virology, University Hospital Basel, Basel, Switzerland, 13 Collegium Helveticum, Zurich, Switzerland

These authors contributed equally to this work.

¶ Membership of Swiss HIV Cohort Study (SHCS) is provided in Supporting Information file S1 Acknowledgments.

* nadia.neuner-jehle@usz.ch (NNJ), roger.kouyos@uzh.ch (RDK)

Abstract

The pathogenesis of HIV-1 infection is governed by a highly dynamic, time-dependent interaction between the host and the viral genome. In this study, we developed a novel systematic approach to assess the host-virus interaction, using average pairwise viral diversity as a proxy for time since infection, and applied this method to nearly whole viral genome sequences (n = 4,464), human leukocyte antigen (HLA) genotyping data (n = 1,044), and viral RNA load (VL) measurements during the untreated chronic phase (n = 829) of Swiss HIV Cohort Study participants. Our systematic genome-wide screen revealed for 98 HLA/ viral-variant pairs a signature of immune-driven selection in the form of an HLA-dependent effect of infection time on the presence of HIV amino acid variants. Of these pairs, 12 were found to have an effect on VL. Furthermore, 28/58 pairs were validated by time-to-event analyses and 48/92 by computational HLA-epitope predictions. Our diversity-based approach allows a powerful and systematic investigation of the interaction between the virus and cellular immunity, revealing a notable subset of such interaction effects. From an evolutionary perspective, these observations underscore the complexity of HLA-mediated selection pressures on the virus that shape viral evolution and pathogenesis. shared publicly due to the sensitive nature and privacy concerns (see https://www.shcs.ch/294open-data-statement-shcs). Investigators with a request for selected data can send a proposal to the Swiss HIV Cohort Study (www.shcs.ch/ contact). The provision of data will be evaluated by the Scientific Board of the Swiss HIV Cohort Study and the study team and will be subject to Swiss legal and ethical regulations. The main coding scripts for this study will be made available in a public GitHub repository (under https://github.com/ nneune/HLA_APD).

Funding: This study has been financed within the framework of the Swiss HIV Cohort Study (SHCS), which is supported by the Swiss National Science Foundation (SNSF; grant #201369), by SHCS project #910, and by the SHCS research foundation. The data are gathered by the Five Swiss University Hospitals, two Cantonal Hospitals, 15 affiliated hospitals, and 36 private physicians (listed in http://www.shcs.ch/180-health-careproviders). This work was further supported by the SNSF (grant numbers 177499 to H.F.G. in the framework of the SHCS; 179571 and 141067 to H. F.G.; and 207957 and 155851 to R.D.K.); the Yvonne-Jacob Foundation (to H.F.G.); the University of Zurich Clinical Research Priority Program for Viral Infectious Disease, the Zurich Primary HIV Infection Cohort Study (to H.F.G.); and an unrestricted research grant from Gilead Sciences (to the SHCS Research Foundation). N. N.-J.'s salary was financed in part by grants from the SNSF (207957) and the SHCS (project #910). C.P. was supported by a fellowship from the Collegium Helveticum. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

Competing interests: I have read the journal's policy and the authors of this manuscript have the following competing interests: K.J.M. has received travel grants and honoraria from Gilead Sciences, Roche Diagnostics, GlaxoSmithKline, Merck Sharp & Dohme, Bristol-Myers Squibb, ViiV, and Abbott; and the University of Zurich has received research grants from Gilead Science, Novartis, Roche, and Merck Sharp & Dohme for studies in which K.J.M. serves as principal investigator, and advisory board honoraria from Gilead Sciences and ViiV. A.R. reports support to his institution for advisory board and/or travel grants from MSD, Gilead Sciences, Pfizer, and Moderna, and an investigator-initiated trial (IIT) grant from Gilead Sciences. All honoraria went to his home institution, not to A.R. personally, and all honoraria were provided outside of the submitted work. M.S. reports advisory board consultations from Gilead, ViiV, MSD, paid to his

Author summary

The intricate interplay between viruses and the human immune system is reflected in dynamic associations between the viral and the human genomes. These often take the form of escape dynamics, in which the virus acquires mutations that allow it to evade immune recognition. We developed a novel viral diversity-based method to screen for such interactions across the viral genome systematically and applied it to a unique dataset of HIV-1 sequences and human leukocyte antigen (HLA) variants. We could identify time-dependent interactions between 98 pairs of HLA and viral variants. Among these pairs, 12 were associated with the concentration of viral RNA, longitudinal time-to-event analyses confirmed 28, and 48 were consistent with computational predictions of viral peptide binding to HLA molecules. Our results highlight how the highly dynamic interaction between the viral genome and the immune system shapes viral evolution, and our approach offers new opportunities to systematically study such interactions from real-world cross-sectional data.

Introduction

The rapid evolution and high diversity of HIV pose challenges to the development of universal vaccines against the virus. Also, HIV disease progression varies significantly among individuals and has been associated with host factors [1-3], and viral evolution [4]. The human genome, specifically the human leukocyte antigen (HLA) genes which encode the major histocompatibility complex (MHC), has emerged as a consistent predictor of HIV disease progression [3,5-9]. HLA class I genes (*A*, *B*, and *C*) have been associated with specific HIV mutations that enable viral immune escape by disrupting antigen processing, epitope-MHC binding, or reducing antigen recognition by T cell receptors [6,9]. These mutations provide the virus a selective advantage in the presence of a matching HLA allele but may be associated with reduced fitness in its absence [9]. Regarding HLA class II, the current evidence is conflicting: studies have computationally predicted CD4⁺ T-cell-driven escape [10], but this could not be confirmed *in vitro* [11]. Another study showed that linkage disequilibrium of HLA alleles may explain associations with class II [12].

Genome-wide association studies have explored HLA-HIV genome interactions in the context of antiretroviral treatment-naïve individuals, finding viral mutations associated with HLA alleles [9,12,13], and identifying "protective" HLA class I alleles (e.g. B*57:01) associated with lower viral load and slower disease progression [3,7,9,12–16].

However, as most of these interactions were found as associations between HIV mutations and HLA alleles in cross-sectional studies, their temporal dynamics remain unclear. Previous longitudinal studies described HIV escape mutations occurring within three years after infection. In the presence of a "protective" HLA allele, mutations occurred more rapidly, and in its absence, they reverted quicker upon transmission [16–20]. These studies, however, focused on specific viral epitopes in small populations, therefore lacking the systematic approach of the genome-wide screens. We developed a novel viral diversity-based approach that allows us to investigate the time-dependent effects of HLA-associated HIV adaptation from cross-sectionally sampled viral genomes.

Based on comprehensive data from the Swiss HIV Cohort Study (SHCS) [21], we applied a previously established and validated viral genetic diversity score (average pairwise diversity, APD [22,23]) as a reliable proxy for time since infection (TSI). Determining the TSI can be

institution, and travel grants for conferences from Gilead, paid to his institution. E.B.'s institution has received research grants from Gilead and Merck unrelated to this work; E.B.'s institution has also received consultancy fees and travel grants from Gilead, ViiV, Merck, Pfizer, Astra Zeneca, Moderna, Abbvie, and Ely Lilly. J.N. has received travel grants from Gilead. H.F.G. has received grants from the Yvonne-Jacob Foundation, the Clinical Research Priority Program of the University of Zurich, and Gilead Sciences, and, outside of this study, grants for unrestricted research from the Swiss HIV Cohort Study, the Swiss National Science Foundation, the National Institutes of Health, the Bill and Melinda Gates Foundation, Gilead, and ViiV; and personal fees as a consultant for Merck, ViiV Healthcare, and Gilead Sciences and as a member of the Data and Safety Monitoring Board for Merck. H.F.G.'s institution has received educational grants unrelated to this work from Gilead, ViiV, MSD, Abbvie, Pfizer, and Sandoz. C.P. has received fellowships from the Collegium Helveticum. R.D.K. has received grants from Gilead Sciences, the National Institutes of Health, the Swiss National Science Foundation, and the Swiss HIV Cohort Study. N.N.J. has received support from the Swiss National Science Foundation and the Swiss HIV Cohort Study. All other authors report no potential conflicts of interest.

challenging as people may live with HIV for many years before being diagnosed [22,23]. HIV diversity increases throughout the infection, and this diversity can be used to derive the TSI. Of several methods that utilize viral diversity, APD was characterized by a mean absolute error of less than one year and higher sensitivity and specificity than, for example, the fraction of ambiguous nucleotides, FAN, method [22,23]. We used APD to systematically assess the time-dependence of HLA/HIV-variant associations. This has allowed us to gain novel insights into the evolving relationship between the human immune system and the ever-adapting HIV.

Results

Study participants

Among all SHCS participants who met our selection criteria (Fig 1A and Table 1), 1,044 had at least one ART-naïve HIV sample sequenced and their HLA alleles genotyped (analysis lb), and 829 had VL measurements within 180 days around the sample date (analysis II). For the longitudinal analysis III, 130 participants had 340 sequences available. Most participants were male (86.2–90%), had "White" self-reported ethnicity (93.1–94.4%), and were of European or Northern American origin (90.0–93.8%). The time between the first HIV diagnosis and ART start varied between analysis lb (median [Interquartile range (IQR)]: 2.15 [0.31, 4.76] years) and analysis II (median [IQR]: 2.93 [1.41, 5.65] years).

HIV amino acid variants associated with HLA alleles

Among the 1,528 SHCS participants included in statistical analysis la, we identified 1,146,205 combinations of HLA alleles and HIV amino acid variants (S1 Fig and S1 Table). We employed a systematic pre-screening to narrow the selection down to HLA/HIV-variant pairs of interest, resulting in the following: First, we identified pairs that yielded a statistical power of at least 80% for detecting an HLA/HIV-variant association with an odds ratio (OR) of three (n = 208,224). Next, we selected those pairs that were associated in a Fisher's exact test (FDR<0.2) (n = 532). Among these 532 associations between HLA alleles and HIV variants, 99 were no longer statistically significant (FDR-corrected p>0.05) when tested in multivariable logistic regression models adjusted for APD and the first ten human genome-based Principal Components (PCs) (S1 Fig and S1 Table). Ancestral and population structure biases likely accounted for these associations. Of the 433 remaining significant pairs, 329 showed a positive association, wherein the viral variant was more prevalent when the HLA allele was present (OR median [IQR]: 2.86 [2.19, 3.83], as shown in S2 Fig).

Most of the HIV variants from these 433 pairs were found in the proteins Pol (n = 131), Nef (n = 89), and Gag (n = 77). Among the HIV proteins considered, Vpr had the highest relative frequency of significant HLA/HIV-variant pairs (with 37 pairs/58,404 possible combinations), followed by Nef (89/165,478), Pol (131/365,496), and Gag (77/233,930). The protein Env had the lowest relative frequency (41/930'696) (S1 Table). Most variants (n = 365) were associated with class I HLA alleles (72 with HLA-A, 174 with HLA-B, and 119 with HLA-C), and the remaining 68 variants with class II HLA alleles (5 with HLA-DPA1, 9 with HLA-DPB1, 14 with HLA-DQA1, 25 with HLA-DQB1 and 15 with HLA-DRB1) (S2 Fig). The strongest HLA/HIV-variant association was found for HLA-B*57:01 and Gag242N (OR = 102.67, 95% CI [47.37, 247.06], p<0.001). Five other HLA alleles (A*01:01, C*06:02, DRB1*07:01, DQA1*02:01, and DQB1*03:03) were associated with Gag242N as well (OR median [IQR]: 3.102 [2.11, 5.37]). For Pol, the strongest association was found between HLA-B*15:01 and 149L (PR 93L) (OR = 7.88, 95% CI [4.71, 13.80], p<0.001), and for Nef the strongest association was HLA-A*11:01 with 92R (OR = 16.17, 95% CI [9.53, 27.69], p<0.001).



Fig 1. Statistical analysis plan with participant selection. A) Measurement data selection of Swiss HIV Cohort Study (SHCS) participants with human leukocyte antigen (HLA) alleles and next-generation sequencing (NGS) data for analysis la & lb, HIV load (VL) data for analysis II, and multiple sequences for analysis III. B) HIV amino acid variants are tested in logistic regressions for their association with HLA alleles and their interaction effect between HLA alleles and average pairwise diversity (APD) (analysis lb). Pairs with significant interaction in analysis lb were further tested for their effect on VL (analysis II), retested in a longitudinal setting (analysis III), and assessed by computational HLA-epitope binding predictions. C) Pairs were selected if power>0.8 and Fisher's exact test p-value<0.2 (FDR-corrected). Statistical analyses performed: [la] Presence of HIV variants as function of HLA alleles. [lb] Presence of HIV variants as function of HLA alleles, APD, and interaction between HLA and APD. [II] VL levels as function of HLA/HIV-variant pairs (identified as significantly associated in [lb]); and [III] longitudinal survival analysis of HLA/HIV-variant pairs identified in [lb]. Only samples from ART-naïve participants were used for the grey-shaded analyses ([lb], [II] and [III]).

https://doi.org/10.1371/journal.ppat.1012385.g001

Evaluating HLA-dependent selection for HIV variants reveals 98 timedependent pairs

To assess the time dependence of HLA selection on HIV variants, we evaluated the interaction between HLA and APD on the odds of observing the viral variant. Among the 1,044 SHCS

	Analysis la	Analysis lb	Analysis	Analysis
Sample type	HLA, HIV seq	HLA, HIV seq	VL, HLA, HIV seq	HLA, HIV seq
Number of SHCS participants	1528	1044	829	130
ART-naïve: n (%)	1044 (68.3)	1044 (100.0)	829 (100.0)	130 (100.0)
Sex: Male (%)	1317 (86.2)	914 (87.5)	719 (86.7)	117 (90.0)
Baseline Age*: median [IQR]	37 [<u>31, 44</u>]	36.5 [<u>30, 43</u>]	36 [30, 42]	36 [30, 41]
Region: n (%)				
Europe & North America	1424 (93.2)	979 (93.8)	772 (93.1)	117 (90.0)
Africa	23 (1.5)	17 (1.6)	15 (1.8)	2 (1.5)
Middle & South America	52 (3.4)	26 (2.5)	24 (2.9)	6 (4.6)
Asia	28 (1.8)	21 (2.0)	17 (2.1)	4 (3.1)
other	1 (0.1)	1 (0.1)	1 (0.1)	1 (0.8)
Ethnicity: n (%)				
White	1443 (94.4)	98 <mark>5 (9</mark> 4.3)	778 (93.8)	121 (93.1)
Black	29 (1.9)	19 (1.8)	18 (2.2)	2 (1.5)
Hispano-American	27 (1.8)	19 (1.8)	16 (1.9)	5 (3.8)
Asian	27 (1.8)	20 (1.9)	17 (2.1)	2 (1.5)
other	2 (0.1)	1 (0.1)	0 (0.0)	0 (0.0)
HIV log ₁₀ RNA: median [IQR]	-	-	4.48 [4.04, 4.95]	-
CD4 ⁺ T cell count: median [IQR]	-		401.0 [309.5, 519.0]	-
proviral DNA: n (%)	484 (31.7)	61 (5.8)	36 (4.3)	1 (0.8)
APD _z score: median [IQR]	0.63 [0.26, 1.23]	0.60 [0.29, 1.14]	0.64 [0.35, 1.19]	0.47 [0.30, 0.82]
Years until ART: median [IQR]	1.65 [0.10, 4.58]	2.15 [0.31, 4.76]	.93 [1.41, 5.65]	2.85 [2.14, 4.67]

Table 1. Characteristics of SHCS participants included. The number of participants used for cross-sectional (I and II) and longitudinal analysis (III) and their characteristics are shown for each analysis, restricted to subtype B.

*Age at the SHCS registration date. Abbreviations: HLA = Human Leukocyte Antigen, HIV seq = HIV sequence, VL = viral load, IQR = Interquartile range, APDz = standardized Average Pairwise Diversity, ART = Antiretroviral therapy, SHCS = Swiss HIV Cohort Study.

https://doi.org/10.1371/journal.ppat.1012385.t001

participants included in statistical analysis lb, we found 98 significant HLA/HIV-variant pairs among all 433 pairs with significant associations in Fisher's exact tests (Fig 2A and S2 Table). Most of these pairs (84/98) had a positive interaction effect between HLA allele and APD (OR IQR [1.71,2.28]), i.e., a decreasing chance of HIV variant occurrence over time was observed in the absence of the HLA allele and an increasing chance was observed in its presence, indicating an HLA-dependent selection pressure on the viral variant (Fig 2D). The amino acid variant Pol432R is an example of this larger increase over time in the presence of HLA-A*03:01. For this pair, the HLA-APD interaction term was positive (OR 3.62, 95% CI [2.09, 6.94], p<0.001) (Fig 2B). In the absence of HLA-A*03:01, Pol432R was less likely to occur with increasing APD (OR 0.69, 95% CI [0.58, 0.81], p<0.001), but in the presence of the allele, this trend was reversed (OR 2.50, 95% CI [1.41, 4.45], p = 0.002) (Fig 2C).

Notably, of the remaining 14/98 pairs with significant HLA-APD interactions, four remained stable in absence of the HLA allele, and ten pairs even had negative interaction terms (Fig 2D and S2 Table). An example for this negative HLA/HIV-variant association and APD-HLA interaction effect is the variant Rev57E, as it was more evident over time in the absence of HLA-B*40:01 (Fig 2B and 2C). For 335 pairs, a significant HLA/HIV-variant association but no significant interaction with APD was estimated. For example, the associated pair Gag242N:B*57:01 (Fig 2B) showed no interaction with APD, suggesting a robust and early selection of Gag242N in HLA-B*57:01-carriers.



Fig 2. Associations between HIV amino acid variants, HLA alleles, and average pairwise diversity interactions. A) Shown are Gag242N: B*57:01 and 98 HLA/HIV-variant pairs with significant average pairwise diversity (APD)-HLA interactions (analysis |β). Odds ratios (OR), sorted

by effect size and HIV protein, are derived from Fisher's exact tests, and colored by HLA genes (class | gray, class || red). Pairs of interest are highlighted (bold label, triangle symbol). B) Multivariable logistic regressions of eight HLA/HIV-variant pairs of interest are shown. The effect of HLA allele, APD, and their interaction on the occurrence of the viral variant is presented as odds ratios with 95% confidence intervals (OR [CI]). Sample sizes are indicated (n). C) Distribution of the pairs Pol432R:A*03:01 (blue) and Rev57E:B*40:01 (red) within the study population and the likelihood of observing the variant in the presence (dark color) or absence (light color) of the HLA allele over APD is shown. Labels indicate the sample sizes for each group. Dashed lines represent the fitted values of linear regressions, while plain lines show data with standard error, D) Effect of APD on presence of viral variant (estimated OR from analysis $|\beta\rangle$ in presence/absence of HLA allele. Coloration according to panel B. Positive interaction effects in yellow quadrant (= expected HLA escape mutation). *All estimates exceeding the third quartile+1.5 were adjusted to this threshold. Standardized APD used in panel B and D.

https://doi.org/10.1371/journal.ppat.1012385.g002

Twelve HLA/HIV-variant pairs affected HIV viral load

In linear regression models, HLA-B*57:01 and B*57:03 were highly associated with lower VL (p < 0.001, S3 Fig) and showed significant interaction effects on VL with viral variants of the proteins Gag, Nef, Pol, and Vif. Assessing those effects among all 98 pairs with significant interaction effects between APD and HLA on the viral variant in analysis II, we found 12 HLA/ HIV-variant pairs in total with a significant interaction effect on VL (IQR regression coefficient (β) [0.43,0.61]) (Figs 3D and 3E and S2 Table). All but one of the twelve pairs showed a positive interaction effect on VL, i.e. the VL was higher in HLA-carriers with the mutation and lower in HLA-carriers without the mutation. For instance, those participants with the HLA-A*03:01 allele had a significantly lower VL than those without it (β = -0.45, 95% CI [-0.81, -0.10], p = 0.013). However, when Pol432R was acquired, VL increased again in HLA-A*03:01-carriers (β = 0.60, 95% CI [0.22, 0.98], p = 0.002) (Fig 3A and 3B). Interestingly, the Gag242N:B*57:01 pair exhibited no HLA-APD interaction effect, yet demonstrated increased VL levels upon mutation, following the trend of the other eleven pairs with positive interactions (Fig 3A and 3E). The pair Rev57E:B*40:01, with a negative interaction effect on VL, showed increased VL levels in the presence of HLA-B*40:01 ($\beta = 0.31, 95\%$ CI [0.10, 0.52], p = 0.003), but upon mutation Rev57E, the VL decreased again ($\beta = -0.53$, 95% CI [-1.00, -0.06], p = 0.027) (Fig 3A and 3C). Overall, we saw that the interaction effects of HLA and APD on the HIV variant corresponded with the HLA/HIV-variant interaction effects on VL ($\rho_{spearman} = 0.94$, p < 0.001 for significant HIV-HLA interactions; S5A Fig).

Time-to-event analyses confirm previous associations of HLA with viral APD

Using longitudinally sampled HIV sequences, we evaluated the impact of HLA alleles on the hazards of HIV variant occurrence in analysis III. We excluded 20 pairs of the 98 previously identified HLA/HIV-variant pairs with a significant HLA-APD interaction (Fig 2B and 3D), that had less than two events (i.e. incident mutation) and 20 pairs that did not have at least one event per exposure group (presence/absence of HLA allele). Cox-proportional hazards-regression models revealed significant differences in the hazard of acquiring the viral variant over time in the presence or absence of the HLA allele for 28 of the remaining 58 pairs (hazard ratio [HR] IQR [7.06, 16.99]). The median time to mutation was shorter in the presence of HLA (median [IQR]: 1.36 [1.04, 1.65] years) compared to the participants without the HLA allele (median [IQR]: 1.44 [1.29, 1.48] years). Overall, the correlation between the APD-HLA interaction ORs and the HR of the longitudinal analysis was moderate ($\rho_{spearman} = 0.48, p < 0.001$; S5B Fig), and the directionality of the significant HRs was consistent with the directionality of the ORs (Fig 4A).

Two examples illustrate the range of patterns (Fig 4B and 4C): Pol-432R developed quickly with an incidence rate of 32.3/100 person-years in participants with HLA-A*03:01, whereas the mutation occurred more slowly in participants with another HLA-A allele (4.75/100



Fig 3. Viral load association with human leukocyte antigen and HIV variant interaction. A) Multivariate linear regressions of eight HLA/HIV-variant pairs of interest (same as in Fig 1B), depicted as their association and interaction effect on viral load (VL). B, C) VL distribution over four groups for pair Pol432R:A*03:01 (B) and Rev57E:B*40:01 (C). The absence of human leukocyte antigen (HLA) or HIV variant is coded as '0' and the presence as '1'. Each participant is represented by one point (mean log10 VL), and violin plots show distribution, and boxplots show distribution and group median and IQR of the

distribution. Coloration according to panel A. D, E) Multiple linear regression estimates of HIV amino acid (AA) variant on VL (D) or HLA allele on VL (E) and HLA/HIV-variant interaction on VL plotted for Gag242N:B*57:01 and all 98 HLA/HIV-variant pairs with significant average pairwise diversity (APD)-HLA interactions (analysis $|\beta\rangle$). The 12 HLA/HIV-variant pairs with a significant interaction effect on viral load (p<0.05) are represented by whole symbols and all non-significant (n.s.) by empty symbols. Shapes indicate HIV protein. Coloration according to panel A. Upper-left section in E (yellow) indicates expected HLA escape mutations.

https://doi.org/10.1371/journal.ppat.1012385.g003

person-years; HR 6.75, 95% CI [1.74, 26.19], p = 0.006) (Fig 4B). On the other hand, Rev57E arose more quickly in the absence of HLA-B*40:01 and not at all in the presence of the allele (Fig 4C).

Computational epitope predictions confirm 48 MHC escape mutations

We used computational MHC class | binding prediction, to assess whether the identified HLA-allele/viral-variants pairs could be explained mechanistically. Among the 92 pairs tested (excluding MHC class II, n = 6), 66 pairs (98 epitopes) were predicted to exhibit at least "weak" binding of at least one non-mutated or mutated 9-mer epitope to the corresponding HLA allele (see <u>S2</u> Table for more information on epitopes). For these, the impact of the mutation on predicted binding was quantified as rank ratios between mutation and consensus (nonmutated) sequence (Fig 5A). Compared to the HLA-APD interaction effects inferred in the analysis I, 48/66 pairs exhibited epitopes with effects in the same direction as the interaction effects, i.e., weaker binding upon mutation for escape mutations (42/48) and vice versa for mutations with a negative HLA*APD interaction effect(6/48), and 18 pairs had epitopes with only divergent effects, i.e., stronger binding upon mutation despite escape mutation characteristic (Figs 5A and S5C). Overall, there was a weak correlation between the APD-HLA interaction effects and rank ratios ($\rho_{spearman} = 0.11$, p = 0.38; S5C Fig). For 56/98 epitopes (41/66 pairs), the mutation did not change the binding category (weak/strong binding) corresponding to the binding ranks (S4 Fig). For 24 (22 pairs) the binding category was predicted weaker upon mutation, e.g., Pol432R:A*03:01 position 9 (still strong binder; Fig 5B), and for 18 epitopes (17 pairs) the binding category was predicted stronger upon mutation, e.g., Rev57E: B*40:01 position 2 (Fig 5C).

Sensitivity analyses

As statistical analysis lb was restricted to HIV-1 subtype B and included proviral sequences, we tested in a sensitivity analysis the impact of subtypes and sequence origin on the HLA-APD interaction for the 98 identified pairs. When excluding the 61 proviral DNA sequences, we observed almost identical results ($\rho_{spearman} = 0.96, p < 0.001; S5D$ Fig). Whilst inferring HLA-APD interaction effects for 320 HIV-1 non-B subtype sequences (see S3 Table), we observed a moderate correlation with the effects for subtype B ($\rho = 0.34$, p = 0.005; S5E Fig) and saw that five pairs retained significance, reflecting potential subtype differences, and limited statistical power for non-B subtypes. When the sequences of all subtypes (B and non-B) were pooled, a strong correlation was observed with the effects for subtype B ($\rho = 0.88$, p < 0.001; S5F Fig). In the pooled analysis, the majority of associations (80 of 98) from the subtype B-only analysis (Ib) were recovered. All 80 pairs exhibited significant HLA-APD interaction effects even after adjusting for the first ten viral PCs, thereby accounting for viral population structure. Lastly, HLA alleles associated with the same viral variant were examined in multivariable logistic regressions to test for the effect of linkage disequilibrium among HLA alleles. In these analyses, none of the class II HLA alleles retained their significant interaction with APD (S6 Fig).



Fig 4. Cumulative hazard of developing an amino acid variant over time concerning the presence of HLA alleles. A) Log odds ratios (OR) of interaction effects between APD and HLA on the viral variant versus the Log Hazard Ratio of the respective pair. Significant HLA/HIV-variant interactions (p<0.05) are represented by whole and non-significant (n.s.) by empty symbols. HLA/HIV-variant pairs of interest (same as in Fig 1B) are highlighted. B, C) Cumulative Hazards of acquiring HIV variant Pol432R (B) or Rev57E (C) in presence (dark blue (B) or dark red (C)) of respective HLA allele versus its absence (light blue (B) or light red (C)), are indicated by the respective lines, confidence intervals are illustrated by the shaded area. Censored events are displayed as vertical lines. P-values are shown as log-rank. "Number at risk" indicates the number of participants with outstanding events after the specified time in years.

https://doi.org/10.1371/journal.ppat.1012385.g004



Fig 5. HLA-HIV peptide binding prediction based on mass spectrometry eluted ligands (EL) rank (%). A) Log odds ratio (OR) of the Average Pairwise Diversity (APD)-HLA interaction versus the Log Ratio of EL Rank of mutation/EL Rank of consensus. Upper-right section (yellow) indicates expected major histocompatibility complex (MHC) escape mutations. Labels describe HLA/HIV-variant pairs of interest (same as in Fig 1B) and the epitope position of mutation (ranging from 1 to 9). B, C) Peptides are 9-mer HIV sequences, including the mutation (dark color; Pol432R/ RT277R (B) or Rev57E (C)) or consensus (light color; Pol432K/ RT277K (B) or Rev57G (C)) at different positions (ranging from 1 to 9). The rank was categorized into strong binder (\leq 0.5%),

weak binder (≤2.0%), and non-binder (>2.0%). Non-binding positions are omitted here. All binding prediction computations are derived from NetMHCpan-4.1.

https://doi.org/10.1371/journal.ppat.1012385.g005

Discussion

Viral and human genetic variants interact in an intricate, highly dynamic way, and viral variants may increase or decrease in frequency during an infection depending on the host genome [3, 4]. Combining large-scale genome-wide screens [6, 7, 12–14] with a study of these adaptive dynamics [16, 17] is challenging as it usually requires longitudinally sampled treatment-naive sequences. This is often prohibitive for large patient populations, especially in the era of immediate treatment initiation. To overcome these limitations, we developed a viral diversity-based approach based on which we quantified the HLA-dependent dynamics of viral variants in over a thousand people with HIV.

This approach stands out by harnessing the potential of cross-sectional data and hence bridging between the within-host and epidemiological perspectives. We identified a significant HLA-dependent effect of infection time (approximated by APD) on the frequency of 98 individual HLA/HIV-variant pairs as a signature for HLA-dependent selection. Of these, 84 exhibited a pattern to be expected of HLA-escape mutations, i.e., a decreasing frequency over infection time in the absence of the HLA allele and an increasing frequency in its presence. We validated and contextualized these results by assessing the dynamics of viral variants in longitudinally sampled SHCS participants and by testing to what extent the identified viral variants had an HLA-dependent impact on VL and disrupted binding to the corresponding HLA alleles.

The time-to-event analysis revealed the robustness of our findings and underscored the limitations of longitudinal data regarding statistical power and scarcity of observed incident viral mutations. About one-third (28/98) of escape mutations found by the APD-based approach could be confirmed in time-to-event analyses and the corresponding effect sizes of the diversity-based and longitudinal analyses were significantly correlated ($\rho_{spearman} = 0.48$, p < 0.001). Almost half (40/98) of the pairs could not be assessed in the analysis because they had too small numbers of incident mutations. Previous studies circumvented the limited power of such longitudinal data by limiting their analyses to specific viral epitopes and/or HLA alleles, potentially missing associations revealed by our systematic screening of the cross-sectional data [16,17,24].

Twelve HLA/HIV-variant pairs had a significant HLA-dependent effect on VL and the magnitude of their effect on VL was strongly correlated with the magnitude of the HLA-dependent selection inferred in the diversity-based screen ($\rho = 0.94$, p < 0.001). Most (11/12) of these interaction effects resulted in increased VL, consistent with previous findings [12,13] and their classification as HLA-escape mutations. The increase in VL can be explained by the fact that HLA alleles such as B*57:01, B*57:03, or C*12:03 lose their "protective" effect (lowering VL) when an escape mutation like Gag242N occurs [12,13,15]. This is especially relevant when the escape mutation has low fitness costs or reversion rates, as we saw for example with Vif33R and B*57:01 (Figs 2D and 3D). At the population level, this could enhance infectivity and pathogenicity [25, 26], similar to what was described in a Japanese study with homogeneous A*24:02 distribution selecting for the escape mutation Nef135F [27].

Of the twelve pairs that significantly interacted with VL, only the two pairs Vif33R:B*57:01 and Pol432R:A*03:01 were previously described as significantly lowering VL [12,13]. For all others, this interaction effect on VL was either not significant (Gag-264K, Pol54L, Pol68P, Pol91D, and Tat32L) [13] or never identified before (Pol9L, Rev57E, and Nef102H).

Conversely, for the newly identified pair Rev57E:B*40:01, we observed an HLA-dependent decrease in VL upon mutation. Rev57E accumulated over time in the absence of B*40:01 but disappeared in its presence, and binding predictions indicated that Rev57E induced a shift from non-binding to strong binding to B*40:01. This suggests that Rev57E increases viral fitness in the absence of B*40:01 but exposes a B*40:01-dependent epitope causing a fitness cost in the presence of B*40:01. These findings offer insights into potential epitopes with high mutational barriers and/or deleterious mutations, which may warrant attention in vaccine design efforts.

The identification of "only" 433 HLA/HIV-variant pairs out of over two million possible combinations may appear modest at first glance. However, it is important to note that it is to be expected that only a minority of positions in the viral genome interact with HLA, due to the selectivity of HLA binding, immuno-dominance, restriction by the proteasome, and T cell receptors, as well as constraints to viral evolution [6, 9]. Moreover, our analysis revealed that even fewer pairs exhibited a time-dependent relationship (98/433), with an even smaller subset demonstrating a significant effect on VL (12/98). It is possible that some of the 433 pairs may be due to spurious associations resulting from population structure and founding effects [28]. However, such confounding by population structure cannot explain the HLA-APD interactions observed in 98 pairs (analysis lb) and the effect of HLA in within-patient survival analyses (analysis III), as these reflect HLA-dependent impacts of time since infection on mutation frequency. This robustness is underlined by the fact that most of these pairs are robust to adjustment for viral population structure (S5F Fig), and hence it represents an additional advantage of the viral diversity-based method established here. In fact, this confounding may be one potential additional reason behind the discrepancy between the number of pairs (n = 335) lost from analysis la to lb.

These findings underscore the complexity of the host-virus interaction landscape, and HIV's ability to adapt and evolve in response to immune pressure. This is particularly evident in the observation that for the majority of variants identified to be under HLA-dependent selection in analysis lb (86/98), the interaction effect between HLA and viral variants on VL was statistically insignificant, and they did not exhibit the anticipated virus-load pattern typically associated with escape mutations. This observation may be attributed to limitations in statistical power. However, it is also consistent with previous analyses indicating that moderate differences in viral fitness may often result in no or only weak variations in VL [13, 29].

Overall, we found a weak correlation between HLA-associated selection and predicted disruption of HLA binding ($\rho = 0.11$, p = 0.38). For 26/58 HLA-viral epitopes, the mutation was predicted to disrupt binding to the HLA allele. For the remaining pairs, e.g., Nef-Y135F: A*24:02, the HLA-dependent selection could not be explained by disruption of HLA binding. Thus, experimental characterizations are necessary to reveal the nature of these escapes, which may operate at different levels such as antigen processing or disruption of T cell receptor recognition.

We identified predominantly HLA/HIV-variant associations within the proteins Gag, Pol, and Nef (consistent with [12]). Despite being rather short proteins, Nef and Vpr exhibited numerous time-dependent associations. Relative to the number of possible combinations, we found for Vpr and Nef even more significant HLA/HIV-variant pairs than for Gag and Pol. Although Nef accumulates many mutations during the course of an HIV infection and these mutations hamper Nef's function in HLA downregulation [30], many of these mutations exhibit a different pattern than typical HLA escape mutations (as seen in Fig 2D). Similar deviations from this "escape pattern" were also observed for a high proportion of variants in Vpr, Rev, and Tat. These mutations appear to have had no or moderate costs on HIV replication, consistent with prior findings in Nef [31], while Gag or Pol were less tolerant to mutations.

Consequently, escape mutations in Gag/Pol are subject to rapid reversion in HLA-unmatched individuals [16]. Alternatively, variants like Rev57G or Nef85L may have been selected in the absence of HLAs and were initially stable escape variants that survived transmission and grad-ually accumulated in the population until becoming the consensus [24].

Even though we had good sequence coverage of Env in our study, the number of HLA-HIV associations was relatively low compared to the number of possible combinations. This can be attributed to several factors. For one, Env is highly variable, with three times more possible HLA/HIV-variant combinations than for example Pol, thus it may have fewer stable and conserved epitopes that can be presented to HLA molecules. Furthermore, much of the selection pressure on Env comes from antibody escape, so the lack of detection of HLA class I-dependent effects could be due to the fact that HLA class I-dependent selection plays a weaker role compared to other HLA class II or HLA-independent mechanisms [32,33]. This may result both in a lower number of HLA-HIV interaction pairs and a lower statistical power to detect them.

Prior research often excluded HLA class II alleles [12,13]. Our observations confirm that HLA class II alleles exert comparatively less immune pressure on HIV, with only six such alleles identified through our time-dependent screen. This supports that CD4⁺ T cell responses may exert less direct influence on viral variants compared to CD8⁺ T cell responses in the context of HIV infections. Consistent with Gabrielaite et al. [12], none of the HIV variants kept significant associations with the interaction between APD and HLA class II when adjusted for linkage disequilibrium between HLA alleles affecting the same viral variants.

A comparison of our findings with those of previous studies and the Los Alamos Immunology database revealed that 32 of the 98 pairs identified through our time-dependent analysis were to our knowledge not previously described (S2 Table) and are not included in [12,13,34,35]. This may be attributed to the fact that our systematic screen had a high sequence coverage across the whole HIV genome, even short genes like *vif*, *vpr*, *tat*, and *rev*. Additionally, we included HLA class II alleles in our analysis. However, it is important to note that these associations should be interpreted with caution, as they may be the result of linkage disequilibrium, as mentioned in the previous paragraph.

Our study bears several limitations. Firstly, although the SHCS comprises individuals from diverse ethnic backgrounds, the majority of participants are of European origin, White ethnicity, and HIV-1 subtype B sequences. Consequently, the generalizability of our results may be limited, as allele frequencies and HIV-1 subtypes can vary among populations [36,37]. Furthermore, the analysis of non-B subtype samples was hindered by small sample sizes, potentially causing us to miss true effects in this population. Nevertheless, all analyses were adjusted for population structures to reduce spurious associations caused by underlying demographic parameters. Secondly, we included plasma and proviral cellular-derived NGS sequences, introducing potential biases, but we show that excluding the proviral sequences did not make a difference. Thirdly, we solely analyzed consensus HIV sequences, disregarding within-host diversity. For future studies, especially those involving longitudinal analyses, conducting investigations of intra-host evolution could offer valuable insights.

In conclusion, we have developed a novel viral diversity-based approach to systematically identify HLA-dependent selection across the HIV genome and map potential escape mutations, and we validate these mutations using longitudinal data and binding predictions. Our approach provides an effective assessment of viral adaptation to their host's genome in the course of infection. As it is based on cross-sectional data, it allows screening for such adaptations in large populations and genome-wide screens. The approaches used here are likely to be applicable to other chronic infections in which host factors exert similar selection pressures driving the emergence of escape mutations and viral adaptation.

Methods

Ethics statement

We included participants enrolled in the Zurich Primary HIV Infection Study, an open-label, non-randomized, observational, monocenter study at the University Hospital Zurich, Switzerland (www.clinicaltrials.gov, ID NCT00537966; approved by the ethical committee (KEK-ZH-Nr. EK-1452) [38-40], and the Swiss HIV Cohort Study (SHCS), an ongoing, multicenter, prospective, longitudinal, observational cohort of people with HIV in Switzerland [21]. The SHCS was approved by the participating centers' local ethical committees: Ethikkommission beider Basel ("Die Ethikkommission beider Basel hat die Dokumente zur Studie zustimmend zur Kenntnis genommen und genehmigt."); Kantonale Ethikkommission Bern (21/88); Comité départemental d'éthique des spécialités médicales et de médecine communautaire et de premier recours, Hôpitaux Universitaires de Genève (01–142); Commission cantonale d'éthique de la recherche sur l'être humain, Canton de Vaud (131/01); Comitato etico cantonale, Repubblica e Cantone Ticino (CE 813); Ethikkommission des Kantons St. Gallen (EKSG 12/003); Kantonale Ethikkommission Zürich (KEK-ZH-NR: EK-793). All participants signed written informed consent [21]. We included participants with genotyped HLA alleles, at least one next-generation HIV sequence (NGS), and ART-naïve virus load. For the statistical analyses I-III, the number of participants and samples were further restricted (see below; Fig 1A).

Human genotyping

Human leukocyte antigen (HLA) alleles and the first ten Principal Components (PC) were derived from human-genome single-nucleotide-polymorphism data for 6,607 SHCS participants genotyped in the setting of previous SHCS genetic studies. Genotyping data was processed and imputed separately for each batch as described previously [41], then merged with PC calculated using PLINK 2.00a5LM [42,43]. Classical HLA alleles at the 4-digit level were imputed separately per genotyping batch using SNP2HLA with the Type 1 Diabetes Genetics Consortium reference panel for samples of European ancestry [44] or HLA-TAPAS for all non-European ancestry participants [45]. For participants genotyped more than once, we retained the sample with the least amount of missing common variants.

HIV sequencing, alignment, and variant calling

HIV-genome sequences were obtained by Illumina MiSeq next-generation sequencing (NGS) (see [46–50] for details) from plasma RNA and proviral DNA (see sensitivity analysis excluding the latter). Sequences were assembled *de-novo* with SmaltAlign [51]. The nucleotide bases were called for positions with a reading depth \geq 20. Consensus sequences were generated and retained if \geq 20% of a gene was covered. For statistical analyses I and II we selected the participant's first sequence. If several sequences were available for this date, we prioritized sequences from plasma viruses and then sequences with the highest genome completeness. Amino acid sequences were generated for each HIV gene using BLAST 2.13.0 [52] to select the respective genes on nucleotide level using a reference panel of 459 HIV sequences from the Los Alamos HIV Sequence Database [53], followed by translation with MACSE 2.05 [54]. For the translation and the multiple sequence alignment of these amino-acid sequences, the HIV-1 HXB2 reference genome (GenBank accession number K03455.1) was used. The nomenclature for HIV amino acid variants presented in this study follows the HXB2 numbering system. The alignment was done separately for each analysis and protein using MAFFT 7.520 [55] and MUSCLE 3.8.31 [56].

For statistical analysis III, participants were included if at least two treatment-naïve (grace period 7 days) sequences were available at different time points at least 30 days apart. Statistical analyses I-III were restricted to HIV-1 subtype B (determined by COMET [57] and REGA [58] from whole-genome nucleotide sequences). HIV-1 non-B subtypes were included in sensitivity analyses.

Average pairwise diversity

To approximate the time since infection, we used the average pairwise viral diversity (APD) calculated from the third codon positions of the *pol* and *gag* genes, as this has been shown to be the most accurate [22]. For variants located in the *gag* gene, we used a *gag*-based APD, while for other variants we used a *pol*-based APD. APD values greater than 0.05 were excluded, as they are typically linked to superinfection [22] and thus unsuitable for assessing infection time. Since the use of APD as a proxy for infection time has been derived and validated solely for ART-naïve sequences, all assessments utilizing APD are limited to sequences obtained prior to ART initiation (participants in statistical analysis lb, see Fig 1).

HIV viral load

Chronic-phase HIV-1 viral load (VL) was calculated based on HIV-1 RNA measurements sampled during untreated chronic infection; i.e., the period from 30 days [59] after HIV diagnosis until 30 days before the first AIDS symptoms [60] or until ART initiation (maximally seven days after). For a given viral sequence, the corresponding VL was calculated as the mean log₁₀-transformed HIV-1 RNA copies/ml plasma over ±180 days around its sampling date.

Statistical analyses

To assess the interplay between HLA, viral amino-acid variants, and VL, we performed three series of analyses (Fig 1C): For analysis I, we systematically screened for HLA-dependent selection of viral variants by multivariable logistic regression models with the presence of HIV variant as the outcome and HLA-APD interaction terms (Fig 1B, lb). We first reduced eligible HLA/HIV-variant combinations to those for which the frequency of HLA and HIV variants yielded a statistical power $\ge 80\%$ for detecting an HLA/HIV-variant association with an odds ratio (OR) of three. We then tested the association between HLA and HIV variant (by Fisher's exact test), adjusted for multiple testing by Benjamini-Hochberg, retaining those pairs with a false discovery rate (FDR) < 0.2 [61]. We chose a liberal FDR to avoid rejecting true pairs, as the selected pairs were further validated. In these pairs, we then used logistic regression models to test the association between the presence of the viral variant (1 = present, 0 = absent), and the HLA-allele (la), APD, HLA-APD interaction (lb), and the first ten human-genome PCs. APD was only added to the models in analysis lb and not la, as the participants needed to be ART-naïve. HLA/HIV-variant pairs with significant HLA-APD interactions (FDR-corrected p<0.05) were considered in analyses II and III. Analysis II evaluated the impact of HLA/HIVvariant interaction on VL (Fig 1B, II) using multivariable linear regressions adjusted for APD and the first ten human PCs. Analysis III assessed if participants with particular HLA alleles acquire the corresponding viral mutations at an increased rate, using longitudinal ART-naïve HIV sequences in Cox-proportional-hazards models (Fig 1B, III). In all statistical analyses I-III, the viral variant was compared to the most prevalent variant at a given position. For instance, a variant corresponding to residue X was compared to all non-X variants at a given position. Statistical analyses were conducted in PLINK 2.0 [42,43], and R 4.3.1 [62].

Several sensitivity analyses were performed. We tested whether proviral sequences biased our results, by repeating analysis lb with proviral sequences previously filtered out. We tested the effect of the HIV-1 subtype, by using all ART-naïve non-B sequences to reexamine the HLA-APD interaction for the pairs identified in the analysis lb (additionally adjusting for the first ten viral PCs generated across the whole viral genome to exclude subtype-dependent effects). We assessed the effect of linkage disequilibrium between HLA alleles, by testing HLA alleles associated with the same variant in multivariable analyses.

Computational epitope prediction

We used epitope prediction implemented in NetMHCpan-4.1 to assess the effect of HIV variants on class I HLA binding for the HLA/HIV-variant pairs identified in analysis lb (Fig 1B). NetMHCpan-4.1 predicts the binding of peptides (9-mer) to MHC class I molecules using neural networks [63]. For a given HLA/HIV-variant pair, we assessed the predicted mass-spectrometry eluted ligands (EL) rank differences between the consensus and the mutated peptide (for all 9-mers containing the viral variant). Viral peptides with a binding rank below 2% or 0.5% were classified as binders or strong binders, as previously described [63]. Only epitopes with at least one binding rank below 2% were included and reported.

Supporting information

S1 Fig. Analysis plan with the number of pairs with significant results included. A) HLA/ HIV-variant pairs were selected if power>0.8 and Fisher's exact test p-value<0.2 (FDR-corrected). Statistical analyses performed: [Ia] Presence of HIV variants as function of HLA alleles. [Ib] Presence of HIV variants as function of HLA alleles, APD, and interaction between HLA and APD. [II] VL levels as function of HLA/HIV-variant pairs (identified as significantly associated in [Ib]); and [III] longitudinal survival analysis of HLA/HIV-variant pairs identified in [Ib]. Only samples from ART-naïve participants were used for the grey-shaded analyses ([Ib], [II] and [III]). **B**) Same structure as panel A, but with the number of pairs indicated that show significance in the statistical test or have significant associations in the analyses. The format is x/y, where x stands for the number of pairs with significant associations (or [†] interaction terms) and y stands for the number of pairs that were included in that particular analysis. For analysis III, 40 pairs were excluded from the analysis due to low sample size and lack of events, explaining the discrepancies between the y number in II (98) and III (58).

(TIFF)

S2 Fig. Pairs with significant associations between HIV amino acid variants and HLA alleles. 433 pairs grouped by HIV genes and arranged by odds ratios (ORs). ORs and 95% confidence intervals derived from Fisher's exact test. Color coding is based on HLA genes, with class I represented in gray and class II represented in red. (TIFF)

S3 Fig. Human leukocyte antigen-wide association analysis on HIV viral load. Associations between human leukocyte antigen (HLA) alleles and viral load are calculated in multivariate linear regressions and are grouped by HLA genes. P-values are shown as $-\log_{10}$ transformation, sample size (n) = 3,676. Two thresholds are depicted: Bonferroni threshold (p = 0.05/n; dotted line) and p = 0.05 (dashed line). (TIFF)

S4 Fig. NetMHCpan-4.1 binding predictions: mutation vs. consensus of HIV variant. The coloration of odds ratios (OR) is derived from the HLA-APD interaction model (analysis lb). Pairs with no change in binding rank (strong, weak, no binding) by NetMHCpan-4.1, have lighter shading.

(TIFF)

S5 Fig. Correlation analyses of the interaction effects between HLA and APD on HIV vs. A) interaction effects of HLA/HIV-variant pairs on VL (analysis II estimates), B) Hazard ratios of occurrence of viral variant in presence/absence of HLA (analysis III cox-proportions test estimates), C) EL rank ratios (EL rank mutation/ EL rank consensus; binding predictions derived from NetMHCpan), **D**) Proviral sequences (n = 61) excluded, estimates of the interaction effects between HLA and APD on HIV, E) Non-B subtypes (A, C, AE, and others) estimates of the interaction effects between HLA and APD on HIV. F) All subtypes (A, B, C, AE, and others) estimates of the interaction effects between HLA and APD on HIV. Dark versus light shading indicates significance/ change on y-axis variables. Spearman's rank correlation ρ , *p*-value, and linear regression lines with 95% confidence intervals were calculated either for each subgroup alone (panel A, C) or for all together (panel B, D, E, F). (TIFF)

S6 Fig. Linkage disequilibrium assessment among HLA alleles that are associated with the same HIV variant. Two multivariable logistic regression models were compared—one from analysis b (triangle form) and the other that includes additional interactions with other HLA alleles with the same viral variant (round form). Error bars reflect confidence intervals for the second model with linkage. Color coding is based on HLA genes, with class I represented in gray and class II represented in red.



S1 Table. An overview of sequence coverage and number of combinations HLA and HIV variants per gene. The sequence coverage shows the number of participants/sequences available per gene, whereas all other columns refer to the different filtering steps. See Methods for more details on procedure. (DOCX)

S2 Table. All 98 pairs with significant interaction effects between HLA and APD on the HIV variant. Odds ratios (OR) for the interaction effect of HLA and APD on the viral variant, estimates (β) of the interaction effect between HLA allele and HIV variant on viral load (VL), Hazard Ratios (HR) of the occurrence of the viral variant in presence/absence of HLA, NetMHCpan EL rank binding predictions before and upon mutation in different positions (pos), and its respective HIV 9-mer consensus epitope. 35 HLA-HIV associations pairs were newly described, all others are listed in the Publication reference list and Los Alamos (LA). All ORs and estimates having *p*-values < 0.05 or changing binding categories are marked in bold. (DOCX)

S3 Table. Characteristics of SHCS participants included. The number of participants used for cross-sectional (I and II) and longitudinal analysis (III) is shown for each analysis, with further information on the percentage of subtype B and non-B and the percentage of ART-naïve participants in the study population.

(DOCX)

S1 Acknowledgements. Membership of Swiss HIV Cohort Study (SHCS). (DOCX)

Acknowledgments

We thank the patients for participating in the Swiss HIV Cohort Study (SHCS), the study nurses, physicians, data managers, and administrative assistants. We thank the BEEHIVE collaboration for providing the sequence data of the SHCS participants included in the BEE-HIVE study.

Author Contributions

- **Conceptualization:** Nadia Neuner-Jehle, Marius Zeeb, Huldrych F. Günthard, Chloé Pasin, Roger D. Kouyos.
- Formal analysis: Nadia Neuner-Jehle, Marius Zeeb, Christian W. Thorball, Paul Frischknecht, Chloé Pasin, Roger D. Kouyos.
- Funding acquisition: Nadia Neuner-Jehle, Chloé Pasin, Roger D. Kouyos.
- **Resources:** Christian W. Thorball, Jacques Fellay, Karin J. Metzner, Paul Frischknecht, Kathrin Neumann, Christine Leeman, Andri Rauch, Marcel Stöckle, Michael Huber, Matthieu Perreau, Enos Bernasconi, Julia Notter, Matthias Hoffmann, Karoline Leuzinger.

Supervision: Chloé Pasin, Roger D. Kouyos.

- Writing original draft: Nadia Neuner-Jehle, Marius Zeeb, Huldrych F. Günthard, Chloé Pasin, Roger D. Kouyos.
- Writing review & editing: Nadia Neuner-Jehle, Marius Zeeb, Christian W. Thorball, Jacques Fellay, Karin J. Metzner, Paul Frischknecht, Kathrin Neumann, Christine Leeman, Andri Rauch, Marcel Stöckle, Michael Huber, Matthieu Perreau, Enos Bernasconi, Julia Notter, Matthias Hoffmann, Karoline Leuzinger, Huldrych F. Günthard, Chloé Pasin, Roger D. Kouyos.

References

- Jia XC, Xia ZH, Shi N, Wang YP, Luo ZX, Yang YL, et al. The factors associated with natural disease progression from HIV to AIDS in the absence of ART, a propensity score matching analysis. Epidemiology and Infection. 2020; 148:1–22. https://doi.org/10.1017/S0950268820000540 PMID: 32089142
- Langford SE, Ananworanich J, Cooper DA. Predictors of disease progression in HIV infection: a review. AIDS Research and Therapy. 2007; 4(1):11. https://doi.org/10.1186/1742-6405-4-11 PMID: 17502001
- Naranbhai V, Carrington M. Host genetic variation and HIV disease: from mapping to mechanism. Immunogenetics. 2017; 69(8–9):489–98. Epub 20170710. https://doi.org/10.1007/s00251-017-1000-z PMID: 28695282; PubMed Central PMCID: PMC5537324.
- Goulder PJ, Walker BD. HIV and HLA class I: an evolving relationship. Immunity. 2012; 37(3):426–40. https://doi.org/10.1016/j.immuni.2012.09.005 PMID: 22999948; PubMed Central PMCID: PMC3966573.
- Rohowsky-Kochan C, Skurnick J, Molinaro D, Louria D. HLA antigens associated with susceptibility/ resistance to HIV-1 infection. Hum Immunol. 1998; 59(12):802–15. https://doi.org/10.1016/s0198-8859 (98)00086-x PMID: 9831136.
- Moore CB, John M, James IR, Christiansen FT, Witt CS, Mallal SA. Evidence of HIV-1 adaptation to HLA-restricted immune responses at a population level. Science. 2002; 296(5572):1439–43. https://doi. org/10.1126/science.1069660 PMID: 12029127.
- Fellay J, Shianna KV, Ge D, Colombo S, Ledergerber B, Weale M, et al. A whole-genome association study of major determinants for host control of HIV-1. Science. 2007; 317(5840):944–7. Epub 20070719. https://doi.org/10.1126/science.1143767 PMID: 17641165; PubMed Central PMCID: PMC1991296.
- Fellay J. Host genetics influences on HIV type-1 disease. Antivir Ther. 2009; 14(6):731–8. https://doi. org/10.3851/IMP1253 PMID: 19812435; PubMed Central PMCID: PMC2851194.
- Carlson JM, Le AQ, Shahid A, Brumme ZL. HIV-1 adaptation to HLA: a window into virus–host immune interactions. Trends in Microbiology. 2015; 23(4):212–24. https://doi.org/10.1016/j.tim.2014.12.008 PMID: 25613992
- Erdmann N, Du VY, Carlson J, Schaefer M, Jureka A, Sterrett S, et al. HLA Class-II Associated HIV Polymorphisms Predict Escape from CD4+ T Cell Responses. PLOS Pathogens. 2015; 11(8): e1005111. https://doi.org/10.1371/journal.ppat.1005111 PMID: 26302050

- Wright JK, Brumme ZL, Julg B, Van Der Stok M, Mncube Z, Gao X, et al. Lack of Association between HLA Class II Alleles and *In Vitro* Replication Capacities of Recombinant Viruses Encoding HIV-1 Subtype C Gag-Protease from Chronically Infected Individuals. Journal of Virology. 2012; 86(2):1273–6. https://doi.org/10.1128/jvi.06533-11 PMID: 22090116
- Gabrielaite M, Bennedbæk M, Zucco AG, Ekenberg C, Murray DD, Kan VL, et al. Human Immunotypes Impose Selection on Viral Genotypes Through Viral Epitope Specificity. J Infect Dis. 2021; 224 (12):2053–63. https://doi.org/10.1093/infdis/jiab253 PMID: 33974707; PubMed Central PMCID: PMC8672757.
- Bartha I, Carlson JM, Brumme CJ, McLaren PJ, Brumme ZL, John M, et al. A genome-to-genome analysis of associations between human genetic variation, HIV-1 sequence diversity, and viral control. Elife. 2013; 2:e01123. Epub 20131029. https://doi.org/10.7554/eLife.01123 PMID: 24171102; PubMed Central PMCID: PMC3807812.
- Fellay J, Ge D, Shianna KV, Colombo S, Ledergerber B, Cirulli ET, et al. Common genetic variation and the control of HIV-1 in humans. PLoS Genet. 2009; 5(12):e1000791. Epub 20091224. https://doi.org/ 10.1371/journal.pgen.1000791 PMID: 20041166; PubMed Central PMCID: PMC2791220.
- Carlson JM, Brumme CJ, Martin E, Listgarten J, Brockman MA, Le AQ, et al. Correlates of protective cellular immunity revealed by analysis of population-level immune escape pathways in HIV-1. J Virol. 2012; 86(24):13202–16. Epub 20121010. https://doi.org/10.1128/JVI.01998-12 PMID: 23055555; PubMed Central PMCID: PMC3503140.
- Duda A, Lee-Turner L, Fox J, Robinson N, Dustan S, Kaye S, et al. HLA-Associated Clinical Progression Correlates with Epitope Reversion Rates in Early Human Immunodeficiency Virus Infection. Journal of Virology. 2009; 83(3):1228–39. https://doi.org/10.1128/JVI.01545-08 PMID: 19019964
- Roberts HE, Hurst J, Robinson N, Brown H, Flanagan P, Vass L, et al. Structured Observations Reveal Slow HIV-1 CTL Escape. PLOS Genetics. 2015; 11(2):e1004914. https://doi.org/10.1371/journal.pgen. 1004914 PMID: 25642847
- Alves E, Al-Kaabi M, Keane NM, Leary S, Almeida CM, Deshpande P, et al. Adaptation to HLA-associated immune pressure over the course of HIV infection and in circulating HIV-1 strains. PLoS Pathog. 2022; 18(12):e1010965. Epub 20221216. https://doi.org/10.1371/journal.ppat.1010965 PMID: 36525463.
- Liu MKP, Hawkins N, Ritchie AJ, Ganusov VV, Whale V, Brackenridge S, et al. Vertical T cell immunodominance and epitope entropy determine HIV-1 escape. Journal of Clinical Investigation. 2012. https:// doi.org/10.1172/JCI65330 PMID: 23221345
- Goonetilleke N, Liu MKP, Salazar-Gonzalez JF, Ferrari G, Giorgi E, Ganusov VV, et al. The first T cell response to transmitted/founder virus contributes to the control of acute viremia in HIV-1 infection. Journal of Experimental Medicine. 2009; 206(6):1253–72. https://doi.org/10.1084/jem.20090365 PMID: 19487423
- Scherrer AU, Traytel A, Braun DL, Calmy A, Battegay M, Cavassini M, et al. Cohort Profile Update: The Swiss HIV Cohort Study (SHCS). Int J Epidemiol. 2022; 51(1):33–4j. <u>https://doi.org/10.1093/ije/ dyab141 PMID: 34363666.</u>
- Carlisle LA, Turk T, Kusejko K, Metzner KJ, Leemann C, Schenkel CD, et al. Viral Diversity Based on Next-Generation Sequencing of HIV-1 Provides Precise Estimates of Infection Recency and Time Since Infection. J Infect Dis. 2019; 220(2):254–65. https://doi.org/10.1093/infdis/jiz094 PMID: 30835266.
- Puller V, Neher R, Albert J. Estimating time of HIV-1 infection from next-generation sequence diversity. PLOS Computational Biology. 2017; 13(10):e1005775. https://doi.org/10.1371/journal.pcbi.1005775 PMID: 28968389
- Leslie A, Kavanagh D, Honeyborne I, Pfafferott K, Edwards C, Pillay T, et al. Transmission and accumulation of CTL escape variants drive negative associations between HIV polymorphisms and HLA. Journal of Experimental Medicine. 2005; 201(6):891–902. <u>https://doi.org/10.1084/jem.20041455</u> PMID: 15781581
- Avila-Rios S, Carlson JM, John M, Mallal S, Brumme ZL. Clinical and evolutionary consequences of HIV adaptation to HLA. Current Opinion in HIV and AIDS. 2019; 14(3):194–204. https://doi.org/10. 1097/coh.000000000000541 PMID: 30925534
- Wymant C, Bezemer D, Blanquart F, Ferretti L, Gall A, Hall M, et al. A highly virulent variant of HIV-1 circulating in the Netherlands. Science. 2022; 375(6580):540–5. Epub 20220203. https://doi.org/10.1126/science.abk1688 PMID: 35113714.
- Furutsuki T, Hosoya N, Kawana-Tachikawa A, Tomizawa M, Odawara T, Goto M, et al. Frequent Transmission of Cytotoxic-T-Lymphocyte Escape Mutants of Human Immunodeficiency Virus Type 1 in the Highly HLA-A24-Positive Japanese Population. Journal of Virology. 2004; 78(16):8437–45. https://doi.org/10.1128/JVI.78.16.8437-8445.2004 PMID: 15280452

- Bhattacharya T, Daniels M, Heckerman D, Foley B, Frahm N, Kadie C, et al. Founder effects in the assessment of HIV polymorphisms and HLA allele associations. Science. 2007; 315(5818):1583–6. https://doi.org/10.1126/science.1131528 PMID: 17363674.
- Bonhoeffer S, Funk GA, Günthard HF, Fischer M, Müller V. Glancing behind virus load variation in HIV-1 infection. Trends in Microbiology. 2003; 11(11):499–504. <u>https://doi.org/10.1016/j.tim.2003.09.002</u> PMID: 14607066
- Sudderuddin H, Kinloch NN, Jin SW, Miller RL, Jones BR, Brumme CJ, et al. Longitudinal within-host evolution of HIV Nef-mediated CD4, HLA and SERINC5 downregulation activity: a case study. Retrovirology. 2020; 17(1). https://doi.org/10.1186/s12977-019-0510-1 PMID: 31918727
- Deacon NJ, Tsykin A, Solomon A, Smith K, Ludford-Menting M, Hooker DJ, et al. Genomic structure of an attenuated quasi species of HIV-1 from a blood transfusion donor and recipients. Science. 1995; 270 (5238):988–91. https://doi.org/10.1126/science.270.5238.988 PMID: 7481804.
- 32. Frost SD, Wrin T, Smith DM, Kosakovsky Pond SL, Liu Y, Paxinos E, et al. Neutralizing antibody responses drive the evolution of human immunodeficiency virus type 1 envelope during recent HIV infection. Proc Natl Acad Sci U S A. 2005; 102(51):18514–9. Epub 20051209. https://doi.org/10.1073/pnas.0504658102 PMID: 16339909; PubMed Central PMCID: PMC1310509.
- Illingworth CJR, Raghwani J, Serwadda D, Sewankambo NK, Robb ML, Eller MA, et al. A de novo approach to inferring within-host fitness effects during untreated HIV-1 infection. PLOS Pathogens. 2020; 16(6):e1008171. https://doi.org/10.1371/journal.ppat.1008171 PMID: 32492061
- Los Alamos Immunology database [cited 2024 04]. Available from: https://www.hiv.lanl.gov/content/ immunology/maps/maps.html.
- Brumme ZL, John M, Carlson JM, Brumme CJ, Chan D, Brockman MA, et al. HLA-associated immune escape pathways in HIV-1 subtype B Gag, Pol and Nef proteins. PLoS One. 2009; 4(8):e6687. Epub 20090819. https://doi.org/10.1371/journal.pone.0006687 PMID: 19690614; PubMed Central PMCID: PMC2723923.
- 36. Gonzalez-Galarza FF, McCabe A, Santos EJMD, Jones J, Takeshita L, Ortega-Rivera ND, et al. Allele frequency net database (AFND) 2020 update: gold-standard data classification, open access genotype data and new query tools. Nucleic Acids Res. 2020; 48(D1):D783–D8. https://doi.org/10.1093/nar/gkz1029 PMID: 31722398; PubMed Central PMCID: PMC7145554.
- Bbosa N, Kaleebu P, Ssemwanga D. HIV subtype diversity worldwide. Curr Opin HIV AIDS. 2019; 14 (3):153–60. https://doi.org/10.1097/COH.0000000000000534 PMID: 30882484.
- Rieder P, Joos B, von Wyl V, Kuster H, Grube C, Leemann C, et al. HIV-1 transmission after cessation of early antiretroviral therapy among men having sex with men. Aids. 2010; 24(8):1177–83. <u>https://doi.org/10.1097/QAD.0b013e328338e4de PMID: 20386427</u>
- Rieder P, Joos B, Scherrer AU, Kuster H, Braun D, Grube C, et al. Characterization of human immunodeficiency virus type 1 (HIV-1) diversity and tropism in 145 patients with primary HIV-1 infection. Clinical infectious diseases. 2011; 53(12):1271–9. https://doi.org/10.1093/cid/cir725 PMID: 21998286
- Gianella S, von Wyl V, Fischer M, Niederoest B, Battegay M, Bernasconi E, et al. Effect of early antiretroviral therapy during primary HIV-1 infection on cell-associated HIV-1 DNA and plasma HIV-1 RNA. Antivir Ther. 2011; 16(4):535–45. https://doi.org/10.3851/imp1776 PMID: 21685541.
- Thorball CW, Oudot-Mellakh T, Ehsan N, Hammer C, Santoni FA, Niay J, et al. Genetic variation near CXCL12 is associated with susceptibility to HIV-related non-Hodgkin lymphoma. Haematologica. 2021; 106(8):2233–41. Epub 20210801. https://doi.org/10.3324/haematol.2020.247023 PMID: 32675224; PubMed Central PMCID: PMC8327743.
- **42.** Price AL, Patterson NJ, Plenge RM, Weinblatt ME, Shadick NA, Reich D. Principal components analysis corrects for stratification in genome-wide association studies. Nat Genet. 2006; 38(8):904–9. Epub 20060723. https://doi.org/10.1038/ng1847 PMID: 16862161.
- Chang CC, Chow CC, Tellier LC, Vattikuti S, Purcell SM, Lee JJ. Second-generation PLINK: rising to the challenge of larger and richer datasets. Gigascience. 2015; 4:7. Epub 20150225. https://doi.org/10. 1186/s13742-015-0047-8 PMID: 25722852; PubMed Central PMCID: PMC4342193.
- 44. Jia X, Han B, Onengut-Gumuscu S, Chen WM, Concannon PJ, Rich SS, et al. Imputing amino acid polymorphisms in human leukocyte antigens. PLoS One. 2013; 8(6):e64683. Epub 20130606. https://doi.org/10.1371/journal.pone.0064683 PMID: 23762245; PubMed Central PMCID: PMC3675122.
- 45. Luo Y, Kanai M, Choi W, Li X, Sakaue S, Yamamoto K, et al. A high-resolution HLA reference panel capturing global population diversity enables multi-ancestry fine-mapping in HIV host response. Nat Genet. 2021; 53(10):1504–16. Epub 20211005. https://doi.org/10.1038/s41588-021-00935-7 PMID: 34611364; PubMed Central PMCID: PMC8959399.
- 46. Giallonardo FD, Töpfer A, Rey M, Prabhakaran S, Duport Y, Leemann C, et al. Full-length haplotype reconstruction to infer the structure of heterogeneous virus populations. Nucleic Acids Res. 2014; 42

(14):e115. Epub 20140627. https://doi.org/10.1093/nar/gku537 PMID: 24972832; PubMed Central PMCID: PMC4132706.

- Blanquart F, Wymant C, Cornelissen M, Gall A, Bakker M, Bezemer D, et al. Viral genetic variation accounts for a third of variability in HIV-1 set-point viral load in Europe. PLOS Biology. 2017; 15(6): e2001855. https://doi.org/10.1371/journal.pbio.2001855 PMID: 28604782
- Wan C, Bachmann N, Mitov V, Blanquart F, Céspedes SP, Turk T, et al. Heritability of the HIV-1 reservoir size and decay under long-term suppressive ART. Nature Communications. 2020;11(1). https://doi.org/10.1038/s41467-020-19198-7 PMID: 33139735
- 49. Jaha B, Schenkel CD, Jörimann L, Huber M, Zaheri M, Neumann K, et al. Prevalence of HIV-1 drug resistance mutations in proviral DNA in the Swiss HIV Cohort Study, a retrospective study from 1995 to 2018. Journal of Antimicrobial Chemotherapy. 2023; 78(9):2323–34. https://doi.org/10.1093/jac/dkad240 PMID: 37545164
- Jörimann L, Tschumi J, Zeeb M, Leemann C, Schenkel CD, Neumann K, et al. Absence of proviral HIV-1 evolution in early treated individuals with HIV switching to dolutegravir monotherapy during 48 weeks. J Infect Dis. 2023. Epub 20230727. https://doi.org/10.1093/infdis/jiad292 PMID: 37498738.
- 51. Zaheri M, Schmutz S, Huber M, Zagordi O. SmaltAlign. 1.1.0 ed. Github2021.
- Camacho C, Coulouris G, Avagyan V, Ma N, Papadopoulos J, Bealer K, et al. BLAST+: architecture and applications. BMC Bioinformatics. 2009; 10(1):421. https://doi.org/10.1186/1471-2105-10-421 PMID: 20003500
- 53. Los Alamos HIV Sequence Database [cited 2023 12/03]. Available from: https://www.hiv.lanl.gov/ content/sequence/NEWALIGN/align.html.
- Ranwez V, Douzery EJP, Cambon C, Chantret N, Delsuc F. MACSE v2: Toolkit for the Alignment of Coding Sequences Accounting for Frameshifts and Stop Codons. Molecular Biology and Evolution. 2018; 35(10):2582–4. https://doi.org/10.1093/molbev/msy159 PMID: 30165589
- 55. Katoh K, Standley DM. MAFFT multiple sequence alignment software version 7: improvements in performance and usability. Mol Biol Evol. 2013; 30(4):772–80. Epub 20130116. https://doi.org/10.1093/molbev/mst010.PMID: 23329690; PubMed Central PMCID: PMC3603318.
- 56. Edgar RC. MUSCLE: multiple sequence alignment with high accuracy and high throughput. Nucleic Acids Research. 2004; 32(5):1792–7. https://doi.org/10.1093/nar/gkh340 PMID: 15034147
- 57. Struck D, Lawyer G, Ternes AM, Schmit JC, Bercoff DP. COMET: adaptive context-based modeling for ultrafast HIV-1 subtype identification. Nucleic Acids Res. 2014; 42(18):e144. Epub 20140812. https:// doi.org/10.1093/nar/gku739 PMID: 25120265; PubMed Central PMCID: PMC4191385.
- Pineda-Peña AC, Faria NR, Imbrechts S, Libin P, Abecasis AB, Deforche K, et al. Automated subtyping of HIV-1 genetic sequences for clinical and surveillance purposes: performance evaluation of the new REGA version 3 and seven other tools. Infect Genet Evol. 2013; 19:337–48. Epub 20130507. <u>https://</u> doi.org/10.1016/j.meegid.2013.04.032 PMID: 23660484.
- Robb ML, Eller LA, Kibuuka H, Rono K, Maganga L, Nitayaphan S, et al. Prospective Study of Acute HIV-1 Infection in Adults in East Africa and Thailand. N Engl J Med. 2016; 374(22):2120–30. Epub 20160518. https://doi.org/10.1056/NEJMoa1508952 PMID: 27192360; PubMed Central PMCID: PMC5111628.
- Castro KG, Ward JW, Slutsker L, Buehler JW, Jaffe HW, Berkelman RL, et al. 1993 Revised Classification System for HIV Infection and Expanded Surveillance Case Definition for AIDS Among Adolescents and Adults. Clinical Infectious Diseases. 1993; 17(4):802–10. https://doi.org/10.1093/clinids/17.4.802
- Nguyen H, Thorball CW, Fellay J, Böni J, Yerly S, Perreau M, et al. Systematic screening of viral and human genetic variation identifies antiretroviral resistance and immune escape link. Elife. 2021;10. Epub 20210601. https://doi.org/10.7554/eLife.67388 PMID: 34061023; PubMed Central PMCID: PMC8169104.
- **62.** R Core Team. R: A language and environment for statistical computing. 4.2.2 ed: R Foundation for Statistical Computing, Vienna, Austria.; 2022.
- Reynisson B, Alvarez B, Paul S, Peters B, Nielsen M. NetMHCpan-4.1 and NetMHCIIpan-4.0: improved predictions of MHC antigen presentation by concurrent motif deconvolution and integration of MS MHC eluted ligand data. Nucleic Acids Res. 2020; 48(W1):W449–w54. https://doi.org/10.1093/nar/gkaa379 PMID: 32406916; PubMed Central PMCID: PMC7319546.